Author’s response to reviews

Title: In vitro and in situ study on characterization and mechanism of the intestinal absorption of 2,3,5,4’-tetrahydroxy-stilbene-2-O-β-D-glucoside

Authors:

Cheng Wang (wangcheng@stu.cdtm.edu.cn)
Yimeng Zhou (342091137@qq.com)
Xiaohong Gong (377103197@qq.com)
Li Zheng (1095148157@qq.com)
Yunxia Li (lyxcdutcm@126.com)

Version: 3 Date: 04 Jan 2020

Author’s response to reviews:

Dear Editors:

Thank you for your letter and for the comments concerning our manuscript entitled “In vitro and in situ study on characterization and mechanism of the intestinal absorption of 2,3,5,4’-tetrahydroxy-stilbene-2-O-β-D-glucoside” (PHAT-D-19-00231R2). Those comments are all valuable for revising our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the editor’s comments are as flowing:

1. Authors’ contributions

Please note that based on the current authors’ contributions section, YZ, LZ, YL do not automatically qualify for authorship. Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group, alone, does not usually justify authorship.

A: Thank you very much for your suggestion. We provide more information about the author's personal contributions, and all authors have read and approved the final version.

2. Clean copy
Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

A: Thank you very much for your suggestion. I submitted a clean manuscript.

We believe that these novel findings will be of interest to the readers of BMC Pharmacology and Toxicology. If you have any queries, please don’t hesitate to contact me at the address below.

Thank you and best regards.

Yours sincerely,

Name: Cheng Wang
Tel: 13882056542
E-mail: wangcheng@stu.cdutcm.edu.cn
Address: School of Pharmacy, Chengdu University of Traditional Chinese Medicine, Chengdu 611137, China